Author's response to reviews

Title: Correlation between Human Papillomavirus Infection and Bladder Transitional Cell Carcinoma

Authors:

Mohammadreza Barghi (h_sasan@hotmail.com)
Amir Hajimohammadmehditarb (h_sasan@hotmail.com)
Seyed Mohammad Mehdi Hosseini Moghaddam (h_sasan@hotmail.com)
Bahram Kazemi (h_sasan@hotmail.com)

Version: 2 Date: 5 September 2005

Author's response to reviews: see over
Dear Dr:

We have revised our manuscript entitled "correlation between human papilomavirus infection and bladder Transitional cell carcinoma" in light of the reviewer's comments.

Changes we have made in our manuscript are as follow:

Page 2:  At line 1: objectives changed to objective
At line 7 the phrase "In the two groups was similar" has been changed to "was similar in the two groups"

Page 3:  At line 10 the word "carrier" has been changed to "carriers"

Page 4:  At line 7 the word "served" has been changed to the phrase "were served"
At line 19 the sentence has been changed to: "transported to molecular biology research center of Shaheed Beheshti University of Medical science under a well-controlled condition"

Page 5:  At lines 6 and 7 "269 bp" and "564 bp" were changed to "269-bp" and "564-bp"
At line 9 "50 µl" was changed to "50-µl"

Page 6:  At line 4 and 5 the word "years" has been transferred to after the parenthesis.
At line 18 the phrase "in patients …." has been changed to "2 patients with HPV6/18 and 1 with…"

Page 7:  At line 12 the word "smoker" has been changed to "smokers"
At line 13 and 14 the word "addict" was changed to "addicted"
At line 19 "of having" has been changed to "to have".

Page 8:  At line 7 the word "cell" is added between transitional and carcinoma.
At line 8 "the" has been eliminated before HPV.
At line 16 the phrase "association of HPV with TCC of bladder" has been changed to "association between HPV and TCC and TCC of bladder".
At the last line the phrase "It sounds likely" is changed to "It seems".

Page 9: At line 3 two cammas have been added to before and after of the phrase "E6 and E7"

At line 13 "via" has been changed to "by".

Page 10: At line 2 the phrase "of loss of quality of DNA" has been changed to "of qualitative loss"

At line 3 the word "suggest" is changed to "suggests".

At line 9 the phrase "in bladder carcinogenesis" has been changed to "and bladder carcinogenesis"

We are looking forward to receiving information about reviewing our manuscript. Thank you.

Sincerely

S.M.M Hosseini Moghaddam